

Do extratropical cyclones impact synopticscale variability of the Arctic Oscillation during cold season?

Article

Published Version

Creative Commons: Attribution 4.0 (CC-BY)

Open Access

Qian, S., Hu, H., Hodges, K. ORCID: https://orcid.org/0000-0003-0894-229X and Yang, X.-Q. (2025) Do extratropical cyclones impact synoptic-scale variability of the Arctic Oscillation during cold season? Geophysical Research Letters, 52 (2). e2024GL112747. ISSN 1944-8007 doi: https://doi.org/10.1029/2024GL112747 Available at https://centaur.reading.ac.uk/118506/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1029/2024GL112747

Publisher: American Geophysical Union

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur



CentAUR

Central Archive at the University of Reading

Reading's research outputs online

ADVANCING EARTH AND SPACE SCIENCES

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2024GL112747

Key Points:

- · The correlation between the AO synoptic variability and the exchange of cyclones between the Arctic and subpolar regions is investigated
- A novel concept of Joint Net Cyclone Flux is derived, which is significantly correlated with the AO synoptic variability
- The quantitative piecewise forcing of the extra-tropical cyclones on the synoptic AO-like geopotential height anomalies is revealed

Supporting Information:

Supporting Information may be found in the online version of this article

Correspondence to:

H. Hu. huhaibo@nju.edu.cn

Citation:

Qian, S., Hu, H., Hodges, K. I., & Yang, X.-Q. (2025). Do extratropical cyclones impact synoptic-scale variability of the Arctic Oscillation during cold season? Geophysical Research Letters, 52, e2024GL112747. https://doi.org/10.1029/ 2024GL112747

Received 26 SEP 2024 Accepted 26 DEC 2024

Author Contributions:

Conceptualization: Haibo Hu Funding acquisition: Haibo Hu Investigation: Shengyi Qian Methodology: Haibo Hu, Kevin I. Hodges, Xiu-Oun Yang Supervision: Haibo Hu Validation: Shengvi Oian Visualization: Shengyi Qian Writing - original draft: Shengyi Qian, Haibo Hu Writing - review & editing: Haibo Hu, Kevin I. Hodges

© 2025. The Author(s). This is an open access article under the

terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Do Extratropical Cyclones Impact Synoptic-Scale Variability of the Arctic Oscillation During Cold Season?

Shengyi Qian¹, Haibo Hu¹, Kevin I. Hodges², and Xiu-Qun Yang¹

¹CMA Key Laboratory for Climate Prediction Studies, School of Atmospheric Sciences, Naniing University, Naniing, China, ²Department of Meteorology, National Centre for Atmospheric Science, University of Reading, Reading, UK

Abstract The Arctic Oscillation (AO) is the most significant mode of sea level pressure (SLP) anomalies in the Northern Hemisphere, exhibiting significant multiple-timescale variability from synoptic to decadal. Using NCEP Climate Forecast System Reanalysis data from 1979 to 2022 during the cold season (November-April), this study identifies the relationship between the number of extra-tropical cyclones entering and exiting the Arctic and the AO synoptic variability. The Joint Net Cyclone Flux (JNCF) is significantly correlated with the spatio-temporal evolution of the synoptic AO and the composites of SLP associated with the JNCF produce AOlike patterns. Subsequent piecewise potential vorticity inversion reveals the impacts of extratropical cyclones on the synoptic-scale AO-like geopotential height anomalies at different altitudes. The effects of extratropical cyclones are more important than Arctic stratospheric PV intrusions. Furthermore, the upper-level dynamic processes among all extratropical cyclone effects dominate the evolution of synoptic-scale AO-like geopotential height anomalies.

Plain Language Summary The Arctic Oscillation (AO) plays an important role in the variability of weather and climate across the entire Northern Hemisphere. This study investigates the correlation between the numbers of extratropical cyclones entering and leaving the Arctic and the AO synoptic variability. The value of the NCF in the North Atlantic region minus that in the North America region is defined as the Joint Net Cyclone Flux (JNCF) which is significantly correlated with the AO synoptic variability with a correlation coefficient of 0.32. The composites of SLP relative to the JNCF index manifest as AO-like anomalies. Piecewise potential vorticity (PV) inversion results further reveal the quantitative forcing of extra-tropical cyclones on the synopticscale AO-like geopotential height anomalies at different altitudes. The effects of extratropical cyclones are more important than Arctic stratospheric PV intrusions. Furthermore, the upper-level dynamic processes among all extratropical cyclone effects dominate the evolution of synoptic-scale AO-like geopotential height anomalies, whereas the mid-troposphere latent heat release contributes little. Interestingly, the effects of the lowertroposphere static stability and baroclinicity on the AO-like synoptic anomalies are completely opposite between the western and eastern parts of the North Atlantic-Arctic sector.

1. Introduction

The Arctic Oscillation (AO) is defined as the first mode of the Empirical Orthogonal Function of extratropical sea level pressure (SLP) $(20^{\circ}-90^{\circ}N)$. Its geopotential height anomaly shows a quasi-barotropic vertical structure spatially extending from the near-surface to the stratosphere (Thompson & Wallace, 1998, 2000; Thompson et al., 2000). Many studies have emphasized that the AO plays a significant role in regulating the weather and climate across the entire Northern Hemisphere, such as the East Asian monsoon (Cheung et al., 2012; Chen et al., 2005, 2019; D. Y. Gong & Ho, 2003; D. Y. Gong et al., 2001, 2011; Huang et al., 2013; Wu & Wang, 2002a, 2002b; L. Wang et al., 2009; L. Wang & Chen, 2010; L. Wang & Lu, 2016), the Indian winter monsoon (Midhuna & Dimri, 2019), Arctic sea ice (Jevrejeva et al., 2003; Rigor et al., 2002; Stroeve, Maslanik, et al., 2011), Eurasian snow cover (Bamzai, 2003), and winter surface air temperatures in the Northern Hemisphere (He & Wang, 2013; Kryjov, 2002; H. J. Park & Ahn, 2016; D. X. Wang et al., 2005). The spatial pattern of the AO is characterized by regional heterogeneity, with a primary maximum value center over the Arctic and opposing maximum value centers over the North Pacific and the North Atlantic. Therefore, it is necessary to take into account the regional heterogeneity of the AO in studies. Some previous studies have questioned the relevance of the AO relative to the North Atlantic Oscillation (NAO) (Ambaum et al., 2001; Deser, 2000; Thompson et al., 2003). Whereas others have suggested that the NAO appears to be part of the AO (Thompson & Wallace, 1998; Wallace, 2000). The





temporal correlation between the AO and the NAO is 0.95 for monthly data and they are nearly indistinguishable in structure (Deser, 2000).

Due to the multiple timescales of the AO, its variability is highly complex. Baldwin and Dunkerton (2001) suggested that the AO originates in the Arctic stratosphere. Wallace (2000) proposed that the interaction between the zonally symmetric flow and mid-to-high latitude eddies affects the interannual variability of the AO, while other mechanisms, including air-sea interaction, control the decadal variability of the AO. Additionally, the AO maybe influenced by various external forcings such as solar activity (Chen & Zhou, 2012; da Silva & Avissar, 2005; Huth et al., 2007), volcanic activity (Qu et al., 2021; Stenchikov et al., 2006), tropical forcing (Jia et al., 2009; Lin et al., 2002; Lu & Pandolfo, 2011; Zhou & Miller, 2005), and Eurasian snow cover (Allen & Zender, 2011; G. Gong et al., 2002, 2003a, 2003b). On interannual timescales, Simmonds et al. (2008) have found that the AO correlates with the frequency of Arctic cyclones. Beyond the low-frequency timescale variations of the AO driven by external forcing, high-frequency timescale events, such as East Asian cold surges (Jeong & Ho, 2005; T. W. Park et al., 2011; Woo et al., 2012), extreme heat events in the U.S. (Higgins et al., 2002; Lim & Schubert, 2011; Wettstein & Mearns, 2002), and extreme precipitation in China (Mao et al., 2011), have also been shown to be closely related to the AO. However, the specific spatiotemporal evolution and mechanisms of the AO on high-frequency timescales remains unclear.

X. D. Zhang et al. (2004) defined all cyclones north of 60°N as Arctic cyclones and divided the Arctic into four regions: the North Pacific, North America, North Atlantic, and Eurasia, to study the climatic characteristics of Arctic cyclones. In winter, Arctic cyclones are less frequent but more intense, while in summer, they occur more frequently but are weaker compared to winter and have a longer life cycle (Sorteberg & Walsh, 2008; Vessey et al., 2020; Zahn et al., 2018; X. D. Zhang et al., 2004). High-latitude cyclones can also undergo explosive development similar to mid-latitude cyclones (Rinke et al., 2017). Numerous studies have highlighted that extratropical cyclone have significant impacts on Arctic sea ice (Aue & Rinke, 2023; Aue et al., 2022; Basu et al., 2019; Boisvert et al., 2016; Finocchio et al., 2020, 2022; Kriegsmann & Brümmer, 2014; Liu & He, 2023; Lukovich et al., 2021; Tian et al., 2022; Webster et al., 2019; Wernli & Papritz, 2018; J. L. Zhang et al., 2013). Additionally, extratropical cyclones have been linked to extreme Arctic wave events (Waseda et al., 2021), increased Arctic precipitation (Oh et al., 2020; Stroeve, Serreze, et al., 2011), Arctic temperature warming (Murto et al., 2022; Rogers & Mosleythompson, 1995), and the Beaufort High (Kenigson & Timmermans, 2021). Extratropical cyclones can have different meridional displacement directions. Cyclones that originate in the midlatitudes and then enter the Arctic transport moisture and heat poleward (Fearon et al., 2021; Sorteberg & Walsh, 2008; Tsopouridis et al., 2021; Tsopouridis et al., 2021a, 2021b; Villamil-Otero et al., 2018; Weijenborg & Spengler, 2020), while cyclones that originate in the Arctic and then move toward the mid-latitudes remove polar moisture and heat. The effects of cyclones with different meridional displacement directions on the Arctic are different. Therefore, when investigating the relationship between extratropical cyclones and the AO, cyclones should be categorized into at least two types: those originating outside the Arctic and entering it, and those generated within the Arctic and leaving it. Furthermore, given the zonal heterogeneity of the AO, it is also important to consider the different paths of cyclones entering and leaving the Arctic (X. D. Zhang et al., 2004).

Inspired by the above discussion, the main scientific questions of this study are as follows: Considering the meridional movement directions of extratropical cyclones and the zonal heterogeneity of the AO, do cyclones entering the Arctic and those leaving the Arctic exert a synergistic influence on the synoptic-scale AO variability during the cold season (November–April)? If the extratropical cyclones do impact the AO synoptic variability, what are the phenomena and mechanisms driving this influence? The structure of the paper is as follows: Section 2 outlines the data and methods used in this study. Section 3 presents the relationship between extratropical cyclones in different regions and the AO synoptic variability, as well as the mechanisms that extratropical cyclones influence on the AO synoptic variability. And Section 4 offers the conclusion and discussion.

2. Data and Methods

The data used in this study are the National Centers for Environmental Prediction (NCEP) Climate Forecast System Reanalysis (CFSR) and its extended version, the Climate Forecast System Version 2 (Saha et al., 2010, 2011). CFSR takes into account the coupling of the atmosphere, ocean, land, and sea ice, and provides high temporal and spatial resolution. The CFSR data utilized in this study cover the cold season (November–April) from 1979 to 2022, with a spatial resolution of $0.5^{\circ} \times 0.5^{\circ}$, a temporal resolution of 6 hr, and 37 vertical levels.

The CFSR data is widely used in the studies of jet streams (Hu et al., 2023), mid-latitude cyclones (Qian et al., 2024), subsurface ocean (Hu et al., 2022), oceanic eddies (Hu et al., 2021) etc. The daily AO index (AOI) is obtained from the National Oceanic and Atmospheric Administration (NOAA) Climate Prediction Center. For the daily AOI, the long-term trend is removed and we calculate the 3-day mean to match the cyclone number index. The AO synoptic variability is defined as the 3-day mean of AOI in this study (the results of 3- to 7-day mean are similar). Meanwhile, the long-term trend in each variable used in composite analysis is also removed.

Cyclones are identified and tracked using TRACK (Hodges, 1994, 1995, 1996, 1999) applied to the 6 hourly 850 hPa relative vorticity computed from the NCEP CFSR winds. This method has been widely applied in observational and modeling studies for extratropical cyclone identification (Catto et al., 2010; Hoskins & Hodges, 2002, 2005; Seiler & Zwiers, 2016). Before cyclone identification, the data are preprocessed by spectrally filtering the data to reduce noise and the large-scale background circulation represented by the 0–5 waves is removed. The spectral filtering uses a triangular truncation at 42 waves (T42) and a spectral tapering is used to mitigate the Gibbs phenomenon. A threshold of $1 \times 10^{-5} \text{ s}^{-1}$ is used for initially identifying cyclone feature points as maxima on a polar stereographic projection, these are refined using B-spline interpolation and steepest ascent to find the off-grid locations of the maxima and finally transformed back to the sphere for the tracking. The tracking is performed by first initializing a set of tracks using a nearest neighbor method and then refining these by minimizing a cost function for track smoothness subject to adaptive constraints on displacement distance and track smoothness (Hodges, 1995, 1999) suitable for extra-tropical cyclone motion. Cyclone tracks with lifetimes shorter than 2 days (8 timesteps for 6-hourly data sets) and movement distances less than 10° geodesic (about 1,000 km) are excluded. Additionally, the entire cyclone track including leap days is also excluded to facilitate composite analysis.

In the PV framework, the extratropical cyclone activities can be linked with PV anomalies at different levels from troposphere to stratosphere (Davis & Emanuel, 1991; Reader & Moore, 1995). From PV anomalies in the free atmosphere and potential temperature anomalies at the surface, the three-dimensional geopotential height fields can be retrieved. Piecewise potential vorticity inversion (PPVI) is based on the Ertel potential vorticity (PV) and uses numerical methods to obtain the geopotential height anomaly field from each piecewise PV anomaly field. This method quantitatively estimates the contribution of different forcings represented by each piecewise PV to the changes in the geopotential height (Davis, 1992). Kang and Son (2021) derived a compact linearized PPVI equation, which is simple to calculate and requires fewer computational resources compared to that derived by Davis (1992), with very small errors compared to the reanalysis. This study employs the method proposed by Kang and Son (2021), with the following formula which can be solved by a successive overrelaxation method:

$$q'_n \approx -g \frac{\partial \overline{\theta}}{\partial p} \left[\frac{1}{f} \nabla_p^2 \Phi'_n + \frac{f}{\sigma} \frac{\partial^2 \Phi'_n}{\partial p^2} \right] \tag{1}$$

Where q is the Ertel PV, θ is the potential temperature, Φ is the geopotential, g is the constant of gravity, f is the planetary vorticity, and $\sigma = -\left(\frac{R_d\pi}{p}\right)\left(\frac{\partial\overline{\theta}}{\partial p}\right)$ where R_d is the gas constant of dry air, $\pi = \left(\frac{p}{p_s}\right)^{R_d/C_p}$ is the Exner function, and c_p is the specific heat of dry air under constant pressure. The overbar and prime represent the time mean and climatological anomaly respectively. The subscript *n* denotes the relevant piece of PV. The Dirichlet boundary conditions are used for the horizontal direction and Neumann boundary conditions are used for the vertical direction.

3. Results

According to the dipole pattern of the AO (Thompson & Wallace, 1998), the area north of the 65°N boundary is considered as the Arctic region (Sepp & Jaagus, 2011). A total of 3,156 cyclone tracks originating in the midlatitudes and entering the Arctic are identified, as well as 1,793 cyclone tracks generated in the Arctic and moving toward the mid-latitudes during the cold season from 1979 to 2022 (Figure 1). Similar to the four regions defined by X. D. Zhang et al. (2004), the parallel circle of 65°N is divided into four sectors: the North Pacific (135° E - 135° W), North America (135° W - 55° W), North Atlantic (55° W - 30° E), and Eurasia (30° E - 135° E). The selection of four sectors is sufficient to distinguish the pathway of extratropical cyclones entering and leaving the Arctic, though Sepp and Jaagus (2011) selected more sectors. Moreover, The four sectors

3 of 12



Geophysical Research Letters



Figure 1. The tracks of the (a) total cyclones originating in the mid-latitude and entering the Arctic, and (b) total cyclones originating in the Arctic and moving to the mid-latitude. The blue dots denote the cyclogenesis and the red "×" marks represent the cyclolysis. The black circles are the latitude of 65° N. (c)–(f) The tracks of cyclones entering the Arctic through the four regions respectively. (g)–(j) The tracks of cyclone leaving the Arctic through the four regions respectively. The purple dots denote the position when a cyclone is the most intense. The green, blue, red, and yellow arcs respectively indicate the zonal range of the North Pacific (135° E - 135° W), North Atlantic (55° W - 30° E), and Eurasia (30° E - 135° E) regions. The bold black curves represent the mean track. The number and corresponding percentage of tracks are labeled in the top left of each panel.

are zonally shifted eastward and westward 10° to compare with our result. The results of shifted sectors are similar to the results in this paper indicating that the results are insensitive to the choice of sector boundaries (Figures not shown).

The two oceanic regions of the North Pacific (23%) and North Atlantic (35%) are the main paths for cyclones originating in the mid-latitudes to enter the Arctic. Another set of cyclones enters the Arctic through Eurasia (26%). Cyclones generated in the Arctic primarily leave the Arctic through land regions. The cyclones leaving the Arctic through the North America and Eurasia regions respectively accounts for 38% and 45% of the total number of cyclones leaving the Arctic (Figure 1). In terms of the average intensity represented by maximum vorticity or

minimum SLP anomalies, cyclones in the North Atlantic region are strongest when crossing 65° N, while those in the North Pacific region are the weakest (Figure S1 in Supporting Information S1). Cyclones entering the Arctic through the North Atlantic region are closest to the polar circle when they reach their peak intensity compared to those through other regions. For cyclones leaving the Arctic, a significant portion reach their peak intensity before leaving the Arctic (Figure 1).

To further investigate the impact of extratropical cyclones following different pathways on the AO synoptic variability, we first analyzed the time series of the number of cyclones entering and leaving the Arctic. Since there are many zero values in the daily time series of cyclone number (181 days per cold season), we applied the 3-day accumulation to the daily time series (60 timesteps per cold season). Overall, the correlation between the total number of cyclones entering or leaving the Arctic and the synoptic-scale AOI is very weak. However, interesting results emerge when considering different cyclone pathways. The number of cyclones in the North Pacific and the Eurasia regions are nearly uncorrelated with the synoptic-scale AOI, whereas the number of cyclones in the North America and the North Atlantic regions is significantly correlated with the synoptic-scale AOI (Table S1 in Supporting Information S1).

Since cyclones both entering and leaving the Arctic influence the moisture and heat transport of the Arctic, it is necessary to consider the effects of cyclones entering and leaving the Arctic simultaneously. We define the difference between the number of cyclones entering the Arctic and that leaving the Arctic as the Net Cyclone Flux (NCF). There remains very weak correlation between the total NCF and the synoptic-scale AOI, and the SLP composite results based on the total NCF index further indicating that using the total NCF to discuss the relationship between extratropical cyclones and the AO synoptic variability is not appropriate (Figures 2a and 2d). Notably, the NCF in the North America and North Atlantic regions shows higher significant correlation coefficients of -0.22 and 0.26 with the synoptic-scale AOI, respectively (Table S1 in Supporting Information S1).

We also produced the SLP anomaly composites based on the NCF index in different regions (Figure S2 in Supporting Information S1). The NCF in the North America and North Atlantic regions is significantly correlated with the SLP anomalies over the Arctic. Although the composites of NCF in the Eurasia and the North Pacific regions show significant SLP anomalies, these anomalies are mainly located outside the polar region, and the very low and insignificant correlation coefficient (0.05) with the synoptic-scale AOI suggests that they are unlikely to impact the AO synoptic variability (Table S1 in Supporting Information S1).

From the above analysis, we know that the NCF in the North America and the North Atlantic regions is the key factor correlating with the AO synoptic variability, and they exhibit opposite spatiotemporal correlations with the synoptic-scale AO. Therefore, we define the value of NCF in the North Atlantic region minus that in the North America region as the Joint Net Cyclone Flux (JNCF). The correlation coefficient between the JNCF and the synoptic-scale AOI (0.32) is higher than that between the synoptic-scale AOI and the NCF in either the North America or the North Atlantic region individually (Table S1 in Supporting Information S1).

The composite results associated with the JNCF index show that the JNCF significantly influences the SLP anomalies over the Arctic and North Atlantic, manifesting as synoptic-scale AO-like SLP or geopotential height anomalies (Figures 2b and 2e). The SLP composite associated with the JNCF index is approximately 50% of the intensity of the SLP composite based on the synoptic-scale AOI (Figure 2). Moreover, the AO pattern is commonly accompanied by significant temperature and wind anomalies in the Arctic region, especially in the North Atlantic (Figure S3 in Supporting Information S1). The performances of composite temperature and wind anomaly based on the JNCF index are similar to the SLP anomaly, which show AO-like patterns and reach about 50% intensity of the patterns based on the synoptic-scale AO-like geopotential height anomalies? The method of PPVI is applied to answer this question.

The activities of extratropical cyclones are commonly accompanied by positive PV anomalies in free atmosphere over the cyclone (Reader & Moore, 1995). The PV anomalies in the upper, middle, and lower troposphere are considered to be associated with the stratospheric intrusion caused by dynamical processes, the latent heat release, and the surface baroclinicity respectively (Davis, 1992; Kang & Son, 2021; Seiler, 2019). However, positive PV anomalies also exist to the north of jet stream accompanied by the cyclone activity. It is inaccurate to assume all the PV anomalies in the upper-level atmosphere are caused by stratospheric intrusions, which act as a precursor signal and have significantly influence on the variability of AO (Baldwin & Dunkerton, 2001). The PV anomalies



Geophysical Research Letters



Figure 2. Composites of SLP anomalies (shadings, unit: hPa) and 850-hPa geopotential height anomalies (contours, unit: gpm). (a) The composites during the positive period of total Net Cyclone Flux index (greater than 1.0). (b) The composites during the positive period of Joint Net Cyclone Flux index (greater than 1.0). (d)–(e) are the same as (a)–(b) but for the negative period (less than -1.0). (g) and (h) are the composites of positive and negative AO phases respectively. Only the SLP anomalies with significance of 95% confidence level are drawn.

in the upper-level atmosphere should be separated into at least those associated with the jet stream and those caused by stratospheric intrusion (Qian et al., 2024). Following Qian et al. (2024), we divide the PV into four pieces according to different pressure levels. These are the lower tropospheric PV (1,000, 950, 900, 850 hPa) which represents the static stability and baroclinicity, the middle tropospheric PV (800, 700, 600, 500 hPa) which represents latent heat release, the upper tropospheric PV (400, 300, 250, 200 hPa) which represents upper-level dynamical processes, and the stratospheric PV (150, 100, 50, 20, 10 hPa) which represents stratospheric intrusion. Based on these PV selections, we perform PPVI for the positive and negative phases of the synoptic-scale AO-like geopotential height anomalies. The inversion results are shown in Figure 3.

The anomaly patterns derived from the full PV inversion are mostly consistent with the reanalysis (Figures 3a, 3b, 3g, and 3h). Given the uneven distribution of the maximum value centers of the synoptic-scale AO-like geopotential height anomalies, we focus primarily on the center located in the North Atlantic-Arctic sector (the black sectorial region in Figure 3). The most significant geopotential height anomaly changes from -6 to 6 days are driven by the effects of extratropical cyclones (red bars in Figures 4a and 4d), whereas the changes driven by the effects of Polar Vortex hardly vary (sky-blue bars in Figures 4a and 4d). The contribution associated with extratropical cyclone exchanges between the Arctic and subpolar regions, including surface baroclinicity, latent heat release, and upper-level dynamical process, is as important or even more important than the arctic stratospheric PV intrusion to the AO synoptic variability. Among all the impacts associated with extratropical cyclones, the upper-level dynamic processes associated with extratropical cyclones dominate the evolution of the synoptic-



Figure 3. The 1,000-hPa geopotential height anomalies obtained from the reanalysis and PPVI (unit: gpm). (a) The composite of 1,000-hPa geopotential height anomalies by CFSR for positive period of JNCF index. (b)–(f) are the 1,000-hPa geopotential height anomalies obtained from the full PV, stratospheric PV, upper tropospheric PV, middle tropospheric PV, and lower tropospheric PV for positive period of JNCF index respectively. (g)–(l) are the same as (a)–(f) but for negative period of JNCF index. The black sector (90° W $- 0^{\circ}$, 55° $- 90^{\circ}$ N) indicates the North Atlantic-Arctic sector and the dashed line is the boundary of western (90° W $- 45^{\circ}$ W, 55° $- 90^{\circ}$ N) and eastern (45° W $- 0^{\circ}$, 55° $- 90^{\circ}$ N) parts of the sector.

scale AO-like geopotential height anomalies. Whereas the contribution from latent heat release in the midtroposphere is negligible (Figures 4b, 4c, 4e, and 4f).

However, the geopotential height anomalies derived from the lower-tropospheric PV inversion exhibit an eastward displacement of the center of the maximum value compared to the reanalysis (Figures 3f and 3l). The eastern anomaly in the North Atlantic-Arctic sector is the same as the reanalysis, while the western anomaly is the opposite of the reanalysis. In the western part, lower-tropospheric stability and baroclinicity suppress the development of synoptic-scale AO-like geopotential height anomalies, while in the eastern part, they promote it. Most of extratropical cyclones enter the Arctic through the North Atlantic region during the positive period of JNCF index but through the North America region during the negative period of JNCF index. The Arctic seesaw effect further causes these opposing impacts (Figure S4 in Supporting Information S1).

4. Conclusion and Discussion

Using high-resolution CFSR data, this study identifies 3,156 cyclone tracks originating in the mid-latitude and entering the Arctic, as well as 1,793 cyclone tracks originating in the Arctic and moving to the mid-latitude during the cold season (November–April) from 1979 to 2022. We first examine the correlation between the number of cyclones and the synoptic-scale AO index (AOI). Results indicate that the total number of cyclones entering and



Geophysical Research Letters



Figure 4. The regional average geopotential height anomalies (units: gpm) of the reanalysis and PPVI. (a) The average geopotential height anomalies of Arctic-Atlantic sector (black sector in Figure 3), (b) The average geopotential height anomalies of western sector, and (c) The average geopotential height anomalies of eastern sector for positive period of JNCF index. (d)–(f) are the same as (a)–(c) but for negative period of JNCF index. The negative value of *x*-axis denotes the days before composites. The gray bars represent the mean of the reanalysis. The black, sky-blue, blue, yellow and red bars indicate the geopotential height anomalies mean retrieved from full PV, stratospheric PV, upper-troposphere PV, mid-troposphere PV, and lower-troposphere PV respectively.

leaving the Arctic is uncorrelated with the synoptic-scale AOI. We then explored the relationship between cyclone numbers in four regions (the North Pacific, North America, North Atlantic, and Eurasian) and the synoptic-scale AOI, yielding interesting results.

The relationship between cyclone number in different pathways and the AO synoptic variability shows significant difference. Cyclone numbers in the North Pacific and Eurasian regions are hardly correlated with the synoptic-scale AOI, whereas those in the North America and North Atlantic regions are significantly correlated with the synoptic-scale AOI. Considering that both cyclones entering and leaving the Arctic influence the moisture and heat transport. We define the net cyclone flux (NCF) as the number of cyclones entering the Arctic minus that leaving the Arctic. The NCF index in the North America region presents a significant negative correlation with the synoptic-scale AOI (-0.22), while the NCF index in the North Atlantic regions shows a significant positive correlation (0.26). Composite sea level pressure (SLP) anomalies based on the NCF indices suggest that NCF in the North Atlantic regions, exhibiting opposing spatiotemporal correlations with the AO synoptic variability, significantly influences the SLP anomalies over the Arctic and North Atlantic.

Building on this, we define the Joint Net Cyclone Flux (JNCF) as the NCF in the North Atlantic region minus that in the North America region. The JNCF shows a higher correlation with the synoptic-scale AOI (0.32). Composite SLP and geopotential height anomaly patterns associated with the JNCF index manifest as synoptic-scale AO-like patterns. To reveal the mechanism that JNCF influences the AO synoptic variability, the method of piecewise potential vorticity inversion (PPVI) is applied. The PPVI results indicate that the contribution associated with extratropical cyclone exchanges between the Arctic and subpolar regions, including surface baroclinicity, latent heat release, and upper-level dynamical process, is as important or even more important than the arctic stratospheric PV intrusion to the AO synoptic variability. Among all the impacts associated with extratropical cyclones, the upper-level dynamic processes associated with extratropical cyclones dominate the evolution of the synopticscale AO-like geopotential height anomalies, whereas the latent heat release in the mid-troposphere contributes little. The geopotential height anomaly centers derived from lower tropospheric PV inversion shift eastward compared to the reanalysis. In the western part of the North Atlantic-Arctic sector, lower-tropospheric instability and baroclinicity suppress the development of synoptic-scale AO-like geopotential height anomalies, while in the eastern part, they promote it. This opposite effect in the lower troposphere is determined by different pathways of extratropical cyclone entering the Arctic during the positive and negative periods of JNCF index.

In this study, the poleward motion of extratropical cyclones is very important for the synoptic AO variability. Tamarin and Kaspi (2016) revealed the importance of the upper-level PV anomaly as well as the crucial role played by latent heat release in the poleward motion of the cyclone. Furthermore, the poleward motion of individual cyclones increases with increasing global mean temperature which intensified the horizontal PV advection and latent heat release (Tamarin & Kaspi, 2017). Yao et al. (2023) found that the diabatic heating at 850 hPa and the horizontal advection by the stationary flow at 500 hPa are the main contributors to the poleward movement of extratropical cyclones. To some extent the exchanges of extratropical cyclones between the Arctic and subpolar regions also influences the interannual variability of the AO. The correlation coefficient between JNCF and the AO index on the interannual timescale is 0.73 (with *p*-value < 0.01), with the annual index of JNCFrelated AO accounting for about 50% of the annual AO index variance (Figure S5 in Supporting Information S1). As mentioned in the introduction, the AO is significantly related to the NAO (H. N. Gong et al., 2018, 2024), we also compute the correlation coefficients between the extratropical cyclone numbers and the NAO index (Table S2 in Supporting Information S1). The relationship between the extratropical cyclone numbers and the NAO is similar to that between the AO, however the NAO shows a higher correlation coefficients with the cyclone numbers than the AO. Since different extratropical cyclones vary in intensity and how much moisture and heat they transport, the JNCF defined in this study only considers frequency, which ignores these cyclone properties. Future research needs to combine the factors such as cyclone intensity and their moisture and heat transports with the cyclones entering and leaving the Arctic.

Data Availability Statement

The reanalysis data of the National Centers for Environmental Prediction (NCEP) Climate Forecast System Reanalysis (CFSR) and its extended version, the Climate Forecast System Version 2 (Saha et al., 2010, 2011) are able to download from https://rda.ucar.edu/. The daily AO index (AOI) is obtained from the National Oceanic and Atmospheric Administration (NOAA) Climate Prediction Center (https://www.cpc.ncep.noaa.gov/).

References

- Allen, R. J., & Zender, C. S. (2011). Forcing of the Arctic Oscillation by Eurasian snow cover. Journal of Climate, 24(24), 6528–6539. https://doi.org/10.1175/2011jcli4157.1
- Ambaum, M. H. P., Hoskins, B. J., & Stephenson, D. B. (2001). Arctic Oscillation or North Atlantic Oscillation? Journal of Climate, 14(16), 3495–3507. https://doi.org/10.1175/1520-0442(2001)014<3495:AOONAO>2.0.CO;2
- Aue, L., & Rinke, A. (2023). Cyclone impacts on sea ice concentration in the Atlantic Arctic Ocean: Annual cycle and recent changes. Geophysical Research Letters, 50(17). https://doi.org/10.1029/2023GL104657
- Aue, L., Vihma, T., Uotila, P., & Rinke, A. (2022). New insights into cyclone impacts on sea ice in the Atlantic sector of the Arctic Ocean in winter. *Geophysical Research Letters*, 49(22). https://doi.org/10.1029/2022GL100051
- Baldwin, M. P., & Dunkerton, T. J. (2001). Stratospheric harbingers of anomalous weather regimes. *Science*, 294(5542), 581–584. https://doi.org/ 10.1126/science.1063315
- Bamzai, A. S. (2003). Relationship between snow cover variability and Arctic oscillation index on a hierarchy of time scales. *International Journal of Climatology*, 23(2), 131–142. https://doi.org/10.1002/joc.854
- Basu, S., Zhang, X. D., & Wang, Z. M. (2019). A modeling investigation of Northern Hemisphere extratropical cyclone activity in spring: The linkage between extreme weather and Arctic sea ice forcing. *Climate*, 7(2), 25. https://doi.org/10.3390/cli7020025
- Boisvert, L. N., Petty, A. A., & Stroeve, J. C. (2016). The impact of the extreme winter 2015/16 Arctic cyclone on the Barents-Kara Seas. *Monthly Weather Review*, 144(11), 4279–4287. https://doi.org/10.1175/Mwr-D-16-0234.1
- Catto, J. L., Shaffrey, L. C., & Hodges, K. I. (2010). Can climate models capture the structure of extratropical cyclones? *Journal of Climate*, 23(7), 1621–1635. https://doi.org/10.1175/2009jcli3318.1
- Chen, W., Wang, L., Feng, J., Wen, Z. P., Ma, T. J., Yang, X. Q., & Wang, C. H. (2019). Recent progress in studies of the variabilities and mechanisms of the East Asian monsoon in a changing climate. Advances in Atmospheric Sciences, 36(9), 887–901. https://doi.org/10.1007/ s00376-019-8230-y

This work was supported by the National Key Program for developing Basic Science (Grants 2022YFE0106600 and 2022YFF0801702), the National Natural Science Foundation of China (Grants 42175060), and the Jiangsu Province Science Foundation (Grant BK20201259). The authors are thankful for the support of the Jiangsu Provincial Innovation Center for Climate Change.

- Chen, W., Yang, S., & Huang, R. H. (2005). Relationship between stationary planetary wave activity and the East Asian winter monsoon. *Journal of Geophysical Research*, 110(D14). https://doi.org/10.1029/2004jd005669
- Chen, W., & Zhou, Q. (2012). Modulation of the Arctic Oscillation and the East Asian winter climate relationships by the 11-year solar cycle. *Advances in Atmospheric Sciences*, 29(2), 217–226. https://doi.org/10.1007/s00376-011-1095-3
- Cheung, H. N., Zhou, W., Mok, H. Y., & Wu, M. C. (2012). Relationship between Ural-Siberian blocking and the East Asian winter monsoon in relation to the Arctic Oscillation and the El Nino-Southern Oscillation. *Journal of Climate*, 25(12), 4242–4257. https://doi.org/10.1175/Jcli-D-11-00225.1
- da Silva, R. R., & Avissar, R. (2005). The impacts of the Luni-Solar oscillation on the Arctic Oscillation. Geophysical Research Letters, 32(22). https://doi.org/10.1029/2005gl023418
- Davis, C. A. (1992). Piecewise potential vorticity inversion. Journal of the Atmospheric Sciences, 49(16), 1397–1411. https://doi.org/10.1175/ 1520-0469(1992)049<1397:Ppvi>2.0.Co;2
- Davis, C. A., & Emanuel, K. A. (1991). Potential vorticity diagnostics of cyclogenesis. Monthly Weather Review, 119(8), 1929–1953. https://doi.org/10.1175/1520-0493(1991)119<1929:Pvdoc>2.0.Co;2
- Deser, C. (2000). On the teleconnectivity of the "Arctic Oscillation". Geophysical Research Letters, 27(6), 779–782. https://doi.org/10.1029/1999g1010945
- Fearon, M. G., Doyle, J. D., Ryglicki, D. R., Finocchio, P. M., & Sprenger, M. (2021). The role of cyclones in moisture transport into the Arctic. *Geophysical Research Letters*, 48(4). https://doi.org/10.1029/2020GL090353
- Finocchio, P. M., Doyle, J. D., & Stern, D. P. (2022). Accelerated sea ice loss from late summer cyclones in the new Arctic. *Journal of Climate*, 35(23), 4151–4169. https://doi.org/10.1175/Jcli-D-22-0315.1
- Finocchio, P. M., Doyle, J. D., Stern, D. P., & Fearon, M. G. (2020). Short-term impacts of Arctic summer cyclones on sea ice extent in the Marginal ice zone. *Geophysical Research Letters*, 47(13). https://doi.org/10.1029/2020GL088338
- Gong, D. Y., & Ho, C. H. (2003). Arctic oscillation signals in the East Asian summer monsoon. Journal of Geophysical Research, 108(D2). https://doi.org/10.1029/2002jd002193
- Gong, D. Y., Wang, S. W., & Zhu, J. H. (2001). East Asian winter monsoon and Arctic Oscillation. *Geophysical Research Letters*, 28(10), 2073–2076. https://doi.org/10.1029/2000gl012311
- Gong, D. Y., Yang, J., Kim, S. J., Gao, Y. Q., Guo, D., Zhou, T. J., & Hu, M. (2011). Spring Arctic Oscillation-East Asian summer monsoon connection through circulation changes over the western North Pacific. *Climate Dynamics*, 37(11–12), 2199–2216. https://doi.org/10.1007/ s00382-011-1041-1
- Gong, G., Entekhabi, D., & Cohen, J. (2002). A large-ensemble model study of the wintertime AO-NAO and the role of interannual snow perturbations. *Journal of Climate*, *15*(23), 3488–3499. https://doi.org/10.1175/1520-0442(2002)015<3488:Alemso>2.0.Co;2
- Gong, G., Entekhabi, D., & Cohen, J. (2003a). Modeled Northern Hemisphere winter climate response to realistic Siberian snow anomalies. *Journal of Climate*, 16(23), 3917–3931. https://doi.org/10.1175/1520-0442(2003)016<3917:Mnhwcr>2.0.Co;2
- Gong, G., Entekhabi, D., & Cohen, J. (2003b). Relative impacts of Siberian and north American snow anomalies on the winter Arctic Oscillation. Geophysical Research Letters, 30(16). https://doi.org/10.1029/2003gl017749
- Gong, H. N., Ma, K. J., Liu, B., Cohen, J., & Wang, L. (2024). Structural fluctuations of the Arctic Oscillation tied to the Atlantic Multidecadal Oscillation. Npj Climate and Atmospheric Science, 7(1), 260. https://doi.org/10.1038/s41612-024-00805-z
- Gong, H. N., Wang, L., Chen, W., & Nath, D. (2018). Multidecadal fluctuation of the wintertime Arctic Oscillation pattern and its implication. Journal of Climate, 31(14), 5595–5608. https://doi.org/10.1175/Jcli-D-17-0530.1
- He, S. P., & Wang, H. J. (2013). Impact of the November/December Arctic Oscillation on the following January temperature in East Asia. Journal of Geophysical Research-Atmospheres, 118(23), 12981–12998. https://doi.org/10.1002/2013jd020525
- Higgins, R. W., Leetmaa, A., & Kousky, V. E. (2002). Relationships between climate variability and winter temperature extremes in the United States. Journal of Climate, 15(13), 1555–1572. https://doi.org/10.1175/1520-0442(2002)015<1555:Rbcvaw>2.0.Co;2
- Hodges, K. I. (1994). A general-method for tracking analysis and its application to meteorological data. *Monthly Weather Review*, 122(11), 2573–2586. https://doi.org/10.1175/1520-0493(1994)122<2573:Agmfta>2.0.Co;2
- Hodges, K. I. (1995). Feature tracking on the unit-sphere. *Monthly Weather Review*, 123(12), 3458–3465. https://doi.org/10.1175/1520-0493 (1995)123<3458:Ftotus>2.0.Co;2
- Hodges, K. I. (1996). Spherical nonparametric estimators applied to the UGAMP model integration for AMIP. Monthly Weather Review, 124(12), 2914–2932. https://doi.org/10.1175/1520-0493(1996)124<2914:Sneatt>2.0.Co;2
- Hodges, K. I. (1999). Adaptive constraints for feature tracking. *Monthly Weather Review*, 127(6), 1362–1373. https://doi.org/10.1175/1520-0493 (1999)127<1362:Acfft>2.0.Co;2
- Hoskins, B. J., & Hodges, K. I. (2002). New perspectives on the Northern Hemisphere winter storm tracks. *Journal of the Atmospheric Sciences*, 59(6), 1041–1061. https://doi.org/10.1175/1520-0469(2002)059<1041:Npotnh>2.0.Co;2
- Hoskins, B. J., & Hodges, K. I. (2005). A new perspective on Southern Hemisphere storm tracks. Journal of Climate, 18(20), 4108–4129. https:// doi.org/10.1175/Jcli3570.1
- Hu, H. B., Chen, W. X., Yang, X. Q., Zhao, Y. H., Bai, H. K., & Mao, K. F. (2022). The mode-water-induced interannual variation of the North Pacific subtropical countercurrent and the corresponding winter atmospheric anomalies. *Geophysical Research Letters*, 49(21). https://doi.org/ 10.1029/2022GL100968
- Hu, H. B., Zhao, Y. H., Yang, X. Q., Jiang, S. Y., Mao, K. F., & Bai, H. K. (2023). The influences of the multi-scale sea surface temperature anomalies in the North Pacific on the jet stream in winter. *Journal of Geophysical Research-Atmospheres*, 128(9). https://doi.org/10.1029/ 2022JD038036
- Hu, H. B., Zhao, Y. H., Zhang, N., Bai, H. K., & Chen, F. F. (2021). Local and remote forcing effects of oceanic eddies in the subtropical front zone on the mid-latitude atmosphere in winter. *Climate Dynamics*, 57(11–12), 3447–3464. https://doi.org/10.1007/s00382-021-05877-8
- Huang, R. H., Chen, J. L., Wang, L., & Lin, Z. D. (2013). Characteristics, processes, and causes of the spatio-temporal variabilities of the East Asian monsoon system (vol 29, pg 910, 2012). Advances in Atmospheric Sciences, 30(2), 541. https://doi.org/10.1007/s00376-013-0001-6
- Huth, R., Bochnicek, J., & Hejda, P. (2007). The 11-year solar cycle affects the intensity and annularity of the Arctic Oscillation. Journal of Atmospheric and Solar-Terrestrial Physics, 69(9), 1095–1109. https://doi.org/10.1016/j.jastp.2007.03.006
- Jeong, J. H., & Ho, C. H. (2005). Changes in occurrence of cold surges over East Asia in association with Arctic Oscillation. *Geophysical Research Letters*, 32(14). https://doi.org/10.1029/2005gl023024
- Jevrejeva, S., Moore, J. C., & Grinsted, A. (2003). Influence of the arctic oscillation and El Nino-Southern Oscillation (ENSO) on ice conditions in the Baltic Sea: The wavelet approach. *Journal of Geophysical Research*, *108*(D21). https://doi.org/10.1029/2003jd003417
- Jia, X. J., Lin, H., & Derome, J. (2009). The influence of tropical Pacific forcing on the Arctic Oscillation. *Climate Dynamics*, 32(4), 495–509. https://doi.org/10.1007/s00382-008-0401-y

- Kang, J. M., & Son, S. W. (2021). Development processes of the explosive cyclones over the Northwest Pacific: Potential vorticity tendency inversion. *Journal of the Atmospheric Sciences*, 78(6), 1913–1930. https://doi.org/10.1175/Jas-D-20-0151.1
- Kenigson, J. S., & Timmermans, M. L. (2021). Arctic cyclone activity and the Beaufort high. Journal of Climate, 34(10), 4119–4127. https://doi. org/10.1175/Jcli-D-20-0771.1
- Kriegsmann, A., & Brümmer, B. (2014). Cyclone impact on sea ice in the central Arctic Ocean: A statistical study. *The Cryosphere*, 8(1), 303–317. https://doi.org/10.5194/tc-8-303-2014
- Kryjov, V. N. (2002). The influence of the winter Arctic oscillation on the northern Russia spring temperature. International Journal of Climatology, 22(7), 779–785. https://doi.org/10.1002/joc.746
- Lim, Y. K., & Schubert, S. D. (2011). The impact of ENSO and the Arctic Oscillation on winter temperature extremes in the southeast United States. *Geophysical Research Letters*, 38(15). https://doi.org/10.1029/2011gl048283
- Lin, H., Derome, J., Greatbatch, R. J., Peterson, K. A., & Lu, J. (2002). Tropical links of the Arctic Oscillation. *Geophysical Research Letters*, 29(20). https://doi.org/10.1029/2002gl015822
- Liu, Y. J., & He, Y. J. (2023). Cold season Arctic strong cyclones enhance Atlantification of the Arctic Ocean. *Environmental Research Letters*, 18(11), 114049. https://doi.org/10.1088/1748-9326/ad0518
- Lu, B. W., & Pandolfo, L. (2011). Nonlinear relation of the Arctic oscillation with the quasi-biennial oscillation. *Climate Dynamics*, 36(7–8), 1491–1504. https://doi.org/10.1007/s00382-010-0773-7
- Lukovich, J. V., Stroeve, J. C., Crawford, A., Hamilton, L., Tsamados, M., Heorton, H., & Massonnet, F. (2021). Summer extreme cyclone impacts on Arctic sea ice. *Journal of Climate*, 34(12), 4817–4834. https://doi.org/10.1175/Jcli-D-19-0925.1
- Mao, R., Gong, D. Y., Yang, J., & Bao, J. D. (2011). Linkage between the Arctic Oscillation and winter extreme precipitation over centralsouthern China. *Climate Research*, 50(2–3), 187–201. https://doi.org/10.3354/cr01041
- Midhuna, T. M., & Dimri, A. P. (2019). Impact of arctic oscillation on Indian winter monsoon. *Meteorology and Atmospheric Physics*, 131(4), 1157–1167. https://doi.org/10.1007/s00703-018-0628-z
- Murto, S., Caballero, R., Svensson, G., & Papritz, L. (2022). Interaction between Atlantic cyclones and Eurasian atmospheric blocking drives wintertime warm extremes in the high Arctic. Weather and Climate Dynamics, 3(1), 21–44. https://doi.org/10.5194/wcd-3-21-2022
- Oh, S. G., Sushama, L., & Teufel, B. (2020). Arctic precipitation and surface wind speed associated with cyclones in a changing climate. *Climate Dynamics*, 55(11–12), 3067–3085. https://doi.org/10.1007/s00382-020-05425-w
- Park, H. J., & Ahn, J. B. (2016). Combined effect of the Arctic Oscillation and the Western Pacific pattern on East Asia winter temperature. *Climate Dynamics*, 46(9–10), 3205–3221. https://doi.org/10.1007/s00382-015-2763-2
- Park, T. W., Ho, C. H., & Yang, S. (2011). Relationship between the Arctic Oscillation and cold surges over East Asia. Journal of Climate, 24(1), 68–83. https://doi.org/10.1175/2010jcli3529.1
- Qian, S., Hu, H., Hodges, K. I., Yang, X.-Q., & Song, T. (2024). Synergistic forcing of the troposphere and stratosphere on explosively developing cyclones over the North Pacific during cold season. *Geophysical Research Letters*, 51(17), e2024GL110069. https://doi.org/10.1029/ 2024GL110069
- Qu, W. Z., Huang, F., Zhao, J. P., Du, L., & Cao, Y. (2021). Volcanic activity sparks the Arctic Oscillation. Scientific Reports, 11(1), 15839. https://doi.org/10.1038/s41598-021-94935-6
- Reader, M. C., & Moore, G. W. K. (1995). Stratosphere-troposphere interactions associated with a case of explosive cyclogenesis in the Labrador-Sea. *Tellus Series a-Dynamic Meteorology and Oceanography*, 47(5), 849–863. https://doi.org/10.1034/j.1600-0870.1995.00124.x
- Rigor, I. G., Wallace, J. M., & Colony, R. L. (2002). Response of sea ice to the Arctic oscillation. Journal of Climate, 15(18), 2648–2663. https:// doi.org/10.1175/1520-0442(2002)015<2648:Rositt>2.0.Co;2
- Rinke, A., Maturilli, M., Graham, R. M., Matthes, H., Handorf, D., Cohen, L., et al. (2017). Extreme cyclone events in the Arctic: Wintertime variability and trends. *Environmental Research Letters*, 12(9), 094006. https://doi.org/10.1088/1748-9326/aa7def
- Rogers, J. C., & Mosleythompson, E. (1995). Atlantic arctic cyclones and the Mild Siberian winters of the 1980s. *Geophysical Research Letters*, 22(7), 799–802. https://doi.org/10.1029/95g100301
- Saha, S., Moorthi, S., Pan, H.-L., Wu, X., Wang, J., Nadiga, S., et al. (2010). NCEP climate forecast system reanalysis (CFSR) 6-hourly products, January 1979 to December 2010 (Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory) [Dataset]. https://doi.org/10.5065/D69K487J
- Saha, S., Moorthi, S., Wu, X., Wang, J., Nadiga, S., Tripp, P., et al. (2011). NCEP climate forecast system version 2 (CFSv2) 6-hourly Products (Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory) [Dataset]. https://doi.org/10.5065/D61C1TXF
- Seiler, C. (2019). A climatological assessment of intense extratropical cyclones from the potential vorticity perspective. *Journal of Climate*, 32(8), 2369–2380. https://doi.org/10.1175/Jcli-D-18-0461.1
- Seiler, C., & Zwiers, F. W. (2016). How well do CMIP5 climate models reproduce explosive cyclones in the extratropics of the Northern Hemisphere? *Climate Dynamics*, 46(3-4), 1241–1256. https://doi.org/10.1007/s00382-015-2642-x
- Sepp, M., & Jaagus, J. (2011). Changes in the activity and tracks of Arctic cyclones. Climatic Change, 105(3–4), 577–595. https://doi.org/10.1007/s10584-010-9893-7
- Simmonds, I., Burke, C., & Keay, K. (2008). Arctic climate change as manifest in cyclone behavior. *Journal of Climate*, 21(22), 5777–5796. https://doi.org/10.1175/2008jcli2366.1
- Sorteberg, A., & Walsh, J. E. (2008). Seasonal cyclone variability at 70°N and its impact on moisture transport into the Arctic. *Tellus Series a-Dynamic Meteorology and Oceanography*, 60(3), 570–586. https://doi.org/10.1111/j.1600-0870.2008.00314.x
- Stenchikov, G., Hamilton, K., Stouffer, R. J., Robock, A., Ramaswamy, V., Santer, B., & Graf, H. F. (2006). Arctic Oscillation response to volcanic eruptions in the IPCC AR4 climate models. *Journal of Geophysical Research*, 111(D7). https://doi.org/10.1029/2005jd006286
- Stroeve, J. C., Maslanik, J., Serreze, M. C., Rigor, I., Meier, W., & Fowler, C. (2011). Sea ice response to an extreme negative phase of the Arctic Oscillation during winter 2009/2010. Geophysical Research Letters, 38(2). https://doi.org/10.1029/2010gl045662
- Stroeve, J. C., Serreze, M. C., Barrett, A., & Kindig, D. N. (2011). Attribution of recent changes in autumn cyclone associated precipitation in the Arctic. *Tellus Series a-Dynamic Meteorology and Oceanography*, 63(4), 653–663. https://doi.org/10.1111/j.1600-0870.2011.00515.x
- Tamarin, T., & Kaspi, Y. (2016). The poleward motion of extratropical cyclones from a potential vorticity tendency analysis. *Journal of the Atmospheric Sciences*, 73(4), 1687–1707. https://doi.org/10.1175/Jas-D-15-0168.1
- Tamarin, T., & Kaspi, Y. (2017). The poleward shift of storm tracks under global warming: A Lagrangian perspective. Geophysical Research Letters, 44(20), 10666–10674. https://doi.org/10.1002/2017gl073633
- Thompson, D. W. J., Lee, S., & Baldwin, M. P. (2003). Atmospheric processes governing the Northern Hemisphere annular mode/North Atlantic Oscillation. *Geophysical Monograph Series*, 134, 81–112. https://doi.org/10.1029/134GM05

- Thompson, D. W. J., & Wallace, J. M. (1998). The Arctic Oscillation signature in the wintertime geopotential height and temperature fields. *Geophysical Research Letters*, 25(9), 1297–1300. https://doi.org/10.1029/98g100950
- Thompson, D. W. J., & Wallace, J. M. (2000). Annular modes in the extratropical circulation. Part I: Month-to-month variability. *Journal of Climate*, 13(5), 1000–1016. https://doi.org/10.1175/1520-0442(2000)013<1000:Amitec>2.0.Co;2
- Thompson, D. W. J., Wallace, J. M., & Hegerl, G. C. (2000). Annular modes in the extratropical circulation. Part II: Trends. *Journal of Climate*, 13(5), 1018–1036. https://doi.org/10.1175/1520-0442(2000)013<1018:Amitec>2.0.Co;2
- Tian, Z. X., Liang, X., Zhang, J. L., Bi, H. B., Zhao, F., & Li, C. H. (2022). Thermodynamical and dynamical impacts of an intense cyclone on Arctic sea ice. *Journal of Geophysical Research-Oceans*, 127(12). https://doi.org/10.1029/2022JC018436
- Tsopouridis, L., Spengler, T., & Spensberger, C. (2021). Smoother versus sharper Gulf Stream and Kuroshio sea surface temperature fronts: Effects on cyclones and climatology. *Weather and Climate Dynamics*, 2(4), 953–970. https://doi.org/10.5194/wcd-2-953-2021
- Tsopouridis, L., Spensberger, C., & Spengler, T. (2021a). Characteristics of cyclones following different pathways in the Gulf Stream region. Quarterly Journal of the Royal Meteorological Society, 147(734), 392–407. https://doi.org/10.1002/qj.3924
- Tsopouridis, L., Spensberger, C., & Spengler, T. (2021b). Cyclone intensification in the Kuroshio region and its relation to the sea surface temperature front and upper-level forcing. *Quarterly Journal of the Royal Meteorological Society*, 147(734), 485–500. https://doi.org/10.1002/ qj.3929
- Vessey, A. F., Hodges, K. I., Shaffrey, L. C., & Day, J. J. (2020). An inter-comparison of Arctic synoptic scale storms between four global reanalysis datasets. *Climate Dynamics*, 54(5–6), 2777–2795. https://doi.org/10.1007/s00382-020-05142-4
- Villamil-Otero, G. A., Zhang, J., He, J. X., & Zhang, X. D. (2018). Role of extratropical cyclones in the recently observed increase in poleward moisture transport into the Arctic Ocean. Advances in Atmospheric Sciences, 35(1), 85–94. https://doi.org/10.1007/s00376-017-7116-0
- Wallace, J. M. (2000). North Atlantic oscillatiodannular mode: Two paradigms—One phenomenon. Quarterly Journal of the Royal Meteorological Society, 126(564), 791–805. https://doi.org/10.1002/qj.49712656402
- Wang, D. X., Wang, C. Z., Yang, X. Y., & Lu, J. (2005). Winter Northern Hemisphere surface air temperature variability associated with the Arctic Oscillation and North Atlantic Oscillation. *Geophysical Research Letters*, 32(16). https://doi.org/10.1029/2005gl022952
- Wang, L., & Chen, W. (2010). How well do existing indices measure the strength of the East Asian winter monsoon? Advances in Atmospheric Sciences, 27(4), 855–870. https://doi.org/10.1007/s00376-009-9094-3
- Wang, L., Huang, R. H., Gu, L., Chen, W., & Kang, L. H. (2009). Interdecadal variations of the East Asian winter monsoon and their association with quasi-stationary planetary wave activity. *Journal of Climate*, 22(18), 4860–4872. https://doi.org/10.1175/2009jcli2973.1
- Wang, L., & Lu, M.-M. (2016). The East Asian winter monsoon. In *The global monsoon System* (pp. 51–61). https://doi.org/10.1142/ 9789813200913_0005
- Waseda, T., Nose, T., Kodaira, T., Sasmal, K., & Webb, A. (2021). Climatic trends of extreme wave events caused by Arctic Cyclones in the western Arctic Ocean. *Polar Science*, 27, 100625. https://doi.org/10.1016/j.polar.2020.100625
- Webster, M. A., Parker, C., Boisvert, L., & Kwok, R. (2019). The role of cyclone activity in snow accumulation on Arctic sea ice. Nature Communications, 10(1), 5285. https://doi.org/10.1038/s41467-019-13299-8
- Weijenborg, C., & Spengler, T. (2020). Diabatic heating as a pathway for cyclone clustering encompassing the extreme storm dagmar. Geophysical Research Letters, 47(8). https://doi.org/10.1029/2019GL085777
- Wernli, H., & Papritz, L. (2018). Role of polar anticyclones and mid-latitude cyclones for Arctic summertime sea-ice melting. *Nature Geoscience*, 11(2), 108–113. https://doi.org/10.1038/s41561-017-0041-0
- Wettstein, J. J., & Mearns, L. O. (2002). The influence of the North Atlantic-Arctic Oscillation on mean, variance, and extremes of temperature in the northeastern United States and Canada. *Journal of Climate*, 15(24), 3586–3600. https://doi.org/10.1175/1520-0442(2002)015<3586: Tiotna>2.0.Co;2
- Woo, S. H., Kim, B. M., Jeong, J. H., Kim, S. J., & Lim, G. H. (2012). Decadal changes in surface air temperature variability and cold surge characteristics over northeast Asia and their relation with the Arctic Oscillation for the past three decades (1979–2011). Journal of Geophysical Research, 117(D18). https://doi.org/10.1029/2011jd016929
- Wu, B. Y., & Wang, J. (2002a). Possible impacts of winter Arctic Oscillation on Siberian high, the East Asian winter monsoon and sea-ice extent. Advances in Atmospheric Sciences, 19(2), 297–320. https://doi.org/10.1007/s00376-002-0024-x
- Wu, B. Y., & Wang, J. (2002b). Winter Arctic Oscillation, Siberian high and East Asian winter monsoon. Geophysical Research Letters, 29(19). https://doi.org/10.1029/2002gl015373
- Yao, Y. L., Zhang, Y., Hodges, K. I., & Tamarin-Brodsky, T. (2023). Different propagation mechanisms of deep and shallow wintertime extratropical cyclones over the North Pacific. *Journal of Climate*, 36(23), 8277–8297. https://doi.org/10.1175/Jcli-D-22-0674.1
- Zahn, M., Akperov, M., Rinke, A., Feser, F., & Mokhov, I. I. (2018). Trends of cyclone characteristics in the Arctic and their patterns from different reanalysis data. Journal of Geophysical Research: Atmospheres, 123(5), 2737–2751. https://doi.org/10.1002/2017jd027439
- Zhang, J. L., Lindsay, R., Schweiger, A., & Steele, M. (2013). The impact of an intense summer cyclone on 2012 Arctic sea ice retreat. Geophysical Research Letters, 40(4), 720–726. https://doi.org/10.1002/grl.50190
- Zhang, X. D., Walsh, J. E., Zhang, J., Bhatt, U. S., & Ikeda, M. (2004). Climatology and interannual variability of Arctic cyclone activity: 1948–2002. Journal of Climate, 17(12), 2300–2317. https://doi.org/10.1175/1520-0442(2004)017<2300:Caivoa>2.0.Co;2
- Zhou, S. T., & Miller, A. J. (2005). The interaction of the Madden-Julian oscillation and the Arctic Oscillation. Journal of Climate, 18(1), 143–159. https://doi.org/10.1175/Jcli3251.1